Author's response to reviews

Title: Standard setting: Comparison of two methods

Authors:

Sanju George (sanju.george@talk21.com)
Sayeed M Haque (m.s.haque@bham.ac.uk)
Femi Oyebode (Femi_oyebode@msn.com)

Version: 3 Date: 14 September 2006

Author's response to reviews:

Dera Editor,

We have made the following changes to the format of our manuscript:
1. removed authors' qualifications.
2. Authors' affiliations have been reformatted to in-house style.
3. Section headings and sub-headings are now not in capitals.
4. Table placement in text now removed.
5. Ethical approval section has now been included in the methods section.

Thanking you,

Yours sincerely,

Sanju George